Author's response to reviews

Title: A Wireless Health Outcomes Monitoring System (WHOMS): development and field testing with cancer patients using mobile phones

Authors:

Emilia Bielli (e.bielli@reply.it)
Fabio Carminati (f.carminati@reply.it)
Stella La Capra (s.lacapra@reply.it)
Micaela Lina (psicologia@istitutotumori.mi.it)
Dr Cinzia Brunelli (cinzia.brunelli@istitutotumori.mi.it)
Dr Marcello Tamburini (tamburini@istitutotumori.mi.it)

Version: 5 Date: 5 Jun 2004

re-submitted by Marcello Tamburini with the changes requested by Publisher